



Zekai Ai 🔍, A. Y. C. Nee and S. K. Ong *

Department of Mechanical Engineering, National University of Singapore, Singapore 117575, Singapore; aizekai@u.nus.edu (Z.A.); mpeneeyc@nus.edu.sg (A.Y.C.N.)

* Correspondence: mpeongsk@nus.edu.sg

Abstract: The rapidly increasing adoption of electric vehicles (EVs) globally underscores the urgent need for effective management strategies for end-of-life (EOL) EV batteries. Efficient EOL management is crucial in reducing the ecological footprint of EVs and promoting a circular economy where battery materials are sustainably reused, thereby extending the life cycle of the resources and enhancing overall environmental sustainability. In response to this pressing issue, this review presents a comprehensive analysis of the role of artificial intelligence (AI) in improving the disassembly processes for EV batteries, which is integral to the practical echelon utilization and recycling process. This paper reviews the application of AI techniques in various stages of retired battery disassembly. A significant focus is placed on estimating batteries' state of health (SOH), which is crucial for determining the availability of retired EV batteries. AI-driven methods for planning battery disassembly sequences are examined, revealing potential efficiency gains and cost reductions. AI-driven disassembly operations are discussed, highlighting how AI can streamline processes, improve safety, and reduce environmental hazards. The review concludes with insights into the future integration of electric vehicle battery (EVB) recycling and disassembly, emphasizing the possibility of battery swapping, design for disassembly, and the optimization of charging to prolong battery life and enhance recycling efficiency. This comprehensive analysis underscores the transformative potential of AI in revolutionizing the management of retired EVBs.

Keywords: artificial intelligence; electric vehicle battery disassembly; state-of-health estimation; disassembly sequence planning; disassembly operation

1. Introduction

In recent years, the greenhouse effect has become increasingly severe, and the issue of carbon emissions has become the focus of many countries and individuals. With the popularity of low-carbon and environmental protection initiatives, electric vehicles (EVs) have progressively become a trend to replace fuel vehicles due to their advantages of lower energy consumption when driving the same mileage. According to IEA Global EV Outlook [1], as shown in Figure 1, electric car sales have continued to rise over these years in most parts of the world. However, this also brings new challenges, including recycling lithium batteries. In the production process of EVs, the manufacturing consumption of lithium batteries is enormous. If retired electric vehicle batteries (EVBs) are not recycled, they will cause severe environmental pollution and even risk of fires and explosions. Therefore, the safe and sustainable treatment of retired EVBs is urgent. Currently, the disassembly of lithium batteries in the industry is often destructive and direct, as shown in Figure 2a [2–4]. The main recycling methods are pyrometallurgical recycling [5] and hydrometallurgical recycling [6]. Both recycling methods require a battery to be broken down and sorted first, removing the casing and other non-metallic materials. These two recycling methods can only recover part of the raw materials, and the recycling efficiency is relatively low. Retired EVBs typically retain 80% of their original capacity.



Citation: Ai, Z.; Nee, A.Y.C.; Ong, S.K. Artificial Intelligence in Electric Vehicle Battery Disassembly: A Systematic Review. *Automation* 2024, 5, 484–507. https://doi.org/10.3390/ automation5040028

Academic Editors: Zude Zhou, Quan Liu, Wenjun Xu, F. Javier Ramírez, Marcello Fera, Mario Caterino, Duc Truong Pham and Jeremy Rickli

Received: 2 August 2024 Revised: 17 September 2024 Accepted: 17 September 2024 Published: 24 September 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Scrapping and directly recycling these retired EVBs would result in a significant waste of resources. Dismantling retired EVBs and then recycling the disassembled parts separately can significantly improve recycling efficiency.



Figure 1. Global electric car stock trends, 2010–2023 (adapted from [1]). Notes: BEV = battery electric vehicle; PHEV = plug-in hybrid vehicle.



Figure 2. Life cycle of retired EVBs (adapted from [7]). (a) Direct recycling; (b) echelon utilization.

When an electric vehicle battery's state of health (SOH) is lower than 80%, it must be forcibly retired from EVs. However, such batteries still have specific use values. Currently, there is an optimization solution, namely echelon utilization [8], as shown in Figure 2b. This approach uses a battery's remaining life before it is scrapped, and it can be used in other areas, such as wind farms. Since wind energy has the characteristics of uncertainty in power generation time and power, using these scrapped batteries in wind farms can temporarily save electrical energy. When the health status of these batteries drops to about

50%, it becomes necessary for them to be scrapped [7]. This review will introduce the application of AI to the electric vehicle battery disassembly process.

The current recycling method mainly extracts raw materials, but this method has low returns. In addition, the battery must be shredded first, both in pyrometallurgical recycling and hydrometallurgical recycling. The improper handling of EV batteries may cause a fire and a risk of explosion [9]. In contrast, an efficient method is to disassemble the battery and then recycle it completely. According to the degree of automation, the battery disassembly process can be divided into several categories, namely manual disassembly, semi-automatic disassembly, and fully automated disassembly. Automated disassembly has gradually become a significant trend since there are certain safety risks in the disassembly process. However, the disassembly process is not necessarily the reverse process of the assembly process. Given the different usage situations of EV batteries and different structures from various brands, there may be significant differences between disassembly planning and actual operations, so full automation still needs to be explored in the EV battery disassembly field.

Artificial intelligence (AI) synthesizes computer science, logic, and many other disciplines. AI algorithms simulate human intelligence behaviors to perform tasks, such as decision-making and learning [10,11]. AI has achieved remarkable results in applications, such as image recognition, natural language processing, intelligent robots, etc. Given that AI can help improve the accuracy of detection and automation of disassembly, it is widely used in the disassembly process, such as EVB state-of-health (SOH) estimation and disassembly operations.

Given the significance of retired EVB disassembly and AI potential in this field, summarizing the latest technical advances is highly valuable. This review focuses on the application of AI in the EVB disassembly process, including SOH estimation, disassembly sequence planning, and disassembly operations. To improve the comprehensiveness of this review, the use of AI in other product disassembly processes is also examined, as well as the potential application opportunities of those techniques to EVB disassembly. This survey examines recent research papers on topics related to artificial intelligence, end-of-life EV batteries, state-of-health estimation, disassembly sequence planning, and disassembly operations. Many reviews on similar topics were found during the search, as shown in Table 1. Still, they focused more on the chemical recycling process of lithium batteries or just part processes in battery disassembly. This survey aims to provide a more comprehensive summary of AI applications in the EVB disassembly process. The contributions of this paper are as follows:

- 1. This paper summarizes the current status of electric vehicle batteries' recycling and current issues in the recycling process.
- 2. The applications of AI in the recycling of retired electric vehicle batteries, including SOH estimation, disassembly sequence planning, and disassembly operations, are reviewed.
- 3. Possible future development directions for EVB recycling are discussed.

This article introduces the application of AI to EVB disassembly and the challenges that exist in the EVB disassembly sequence and its steps. The structure of the paper is as follows: Section 2 presents an overview of EVB disassembly problems, including EVB structures and the challenges faced. Section 3 summarizes the AI-based SOH estimation models to assess the remaining useful life of the battery so as to determine whether the battery needs to be fully disassembled. Section 4 discusses disassembly sequence planning to establish the order of the disassembly operations. Section 5 presents AI-driven disassembly operations for EV batteries. Section 6 presents discussions and prospects for EVB disassembly. Section 7 summarizes the key conclusions. A list of abbreviations precedes the references section.

Year	Research focus	Authors
2019	Lithium-ion battery recycling, including pyrometallurgical recovery, physical materials' separation, hydrometallurgical metals' reclamation, direct recycling, and biological metals' reclamation.	Harper et al. [12]
2022	Regulations and new battery directive demand, including current material collection, sorting, transportation, handling, and recycling practices.	Neumann et al. [13]
2022	Artificial intelligence and machine learning applications in EV battery disassembly, including preprocessing, disassembly planning and operation, intelligent interaction and collaboration, and smart design for disassembly.	Meng et al. [14]
2023	LIB recycling methods, including pretreatment, pyrometallurgical recycling, hydrometallurgical recycling, the direct recycling of spent cathode materials, the direct recycling of graphite anode materials, and advanced in situ characterization methods.	Ji et al. [15]
2023	The comprehensive recycling of lithium-ion batteries, including pretreatment, deactivation, dismantling, crushing, and the separation and treatment of electrolytes and solid components.	Yu et al. [16]
2024	Challenges and opportunities for second-life batteries, including battery degradation models, technical assessment procedures, and economic assessment.	Gu et al. [17]
2024	Human–robot collaboration-based EV battery disassembly, including product modeling, disassembly planning, and disassembly operations.	Li et al. [18]
2024	Interpretation from different directions about electric vehicle battery systems' disassembly, including process steps, the level of automation, the use of digital technologies, the level of implementation, and efficiency consideration.	Hertel et al. [19]
2024	A more systematic summary of artificial intelligence applications in electric vehicle battery disassembly, including battery state-of-health detection, disassembly sequence planning, and disassembly operation.	This review

 Table 1. Related reviews about EV battery disassembly and recycling processes.

2. Overview of Electric Vehicle Battery Disassembly Problems and Methodology

2.1. Electric Vehicle Battery Structures

Depending on the dielectric materials used, batteries can be classified into lead-acid, nickel-based, sodium-based, and lithium-based batteries. Lead-acid batteries have stable voltage and a low price but low energy density [20]. They are widely used in uninterruptible power supply systems and backup power supplies. Nickel-based batteries include nickelcadmium batteries and nickel-metal hydride batteries [21]. Nickel-cadmium batteries are commonly used in power tools, such as drills and saws, because of their high discharge rate and long service life. Nickel-metal hydride batteries are also used in some power tools, especially those with high environmental protection requirements. Sodium-ion batteries generally have better thermal stability and safety, reducing the risk of overheating and thermal runaway, but compared to lithium-ion batteries, sodium-ion batteries have a slightly lower energy density [22]. Lithium-ion batteries (LIBs) have the advantages of a high energy density, a high power density, a long life, and no memory effect, so they are widely used in electric vehicles [23]. LIBs are typically formed by an anode, a cathode, an electrolyte, and a separator. The anode and cathode are made of lithium-metal oxide and graphite. LIB packs usually contain a battery module and a battery management system (BMS) [24].

In addition, each module contains multiple tightly packed battery cells to increase energy density and efficiency. Numerous battery modules form a battery pack, which is installed in the chassis of an electric vehicle and used to power a car [25]. The large number of batteries further increases the difficulty of disassembly.

The multiplicity of car manufacturers results in a wide variety of batteries. For example, Tesla has adopted cylindrical-type batteries [26], while Volkswagen uses prismatic battery solutions [27]. Traditional EVBs can be divided into cylindrical, prismatic, and pouch solutions according to the cell's shape [12]. Some new battery structures have emerged

in recent years. BYD introduces a new type of blade battery [28]. The diversity of battery structures poses significant challenges to automated disassembly.

2.2. Challenges in the EVB Disassembly Process

During the EV battery recycling process, the following problems are encountered:

- 1. Low recycling efficiency: In industry, dismantling EVBs is mainly based on destructive dismantling. This method breaks the battery down into smaller parts for further processing through mechanical damage or other means. While this method allows for the rapid separation of battery components, it often damages many valuable components within the battery such that these components cannot be recycled.
- 2. Various types and structures: Different manufacturers use different types of batteries. For example, Tesla uses cylindrical batteries [16], while BYD uses blade batteries [29]. Battery capacity and appearance vary somewhat, even between models from the same vendor. Such a large variety of types and configurations makes automated disassembly difficult.
- 3. Safety risks: End-of-life car batteries contain heavy metals and toxic and hazardous organics, which may release harmful gases during treatment, posing a safety risk to operators.
- 4. High disassembly complexity: The disassembly process has much higher complexity than the assembly process. Many retired EV batteries have rusted screws or even deformed battery structures within them, requiring recognition algorithms to verify the situations; thus, they cannot be disassembled using just the reverse process of the EVB assembly process. Sometimes, the disassembly space is restricted, making it inconvenient for the human and/or robot arm to operate.
- 5. Unpublished data: Most manufacturers do not disclose their vehicles' operating data, which can result in, for example, too little training data for the state-of-health estimation process.

2.3. Methodology

This review adopted the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) [30]. The databases used were Scopus, Web of Science, and Google Scholar. Table 2 shows the keywords used in the search process. As shown in Figure 3, after duplicate papers in different databases were deleted, papers that could not be downloaded or were not relevant to the research topic were excluded. Only papers related to AI applications in EV battery disassembly were retained. These papers mainly include papers on AI in EV battery state-of-health estimation, disassembly sequence planning, and disassembly operations.

Table 2. Search databases and keywords.

Database Keywords		
Scopus	TITLE-ABS-KEY (electric AND vehicle* OR ev*) AND ALL (batter*) AND ALL (disassembl* OR dismant*) AND ALL (artificial AND intelligence OR ai)	
Web of Science	(((AB=(electric vehicle* or ev)) AND AB=(batter*)) AND AB=(artificial intelligence or ai))	
Google Scholar	electric vehicle* OR EV*" AND "batter*" AND "disassembly OR dismantle*" AND "artificial" intelligence OR AI"	



Figure 3. Systematic review results shown in the PRISMA flow diagram.

3. AI-Based Electric Vehicle Battery State-of-Health Estimation

The state of health (SOH) is a crucial indicator of the health of lithium-ion batteries. It reflects the performance and longevity of the battery during its service life, as well as the relative health of the current state compared to the battery's initial state. SOH is usually expressed as a percentage; 100% means the battery is in optimal condition, and 0% means the battery has failed or is close to failure. The direct calculation method of SOH is as shown in Equation (1) [31], where $Q_{max}/mAh =$ the maximum charge available of the battery, and $C_r =$ the rated capacity.

$$\text{SOH}/\% = 100 \frac{Q_{max}}{C_r} \tag{1}$$

EVB SOH evaluates its present performance compared with the fresh state. When the SOH value is lower than 80%, it is regarded as being at its end of life (EOL) and must be obligatorily retired from the electric vehicle. It can be disassembled to the cell level, and then each cell is tested for SOH; cells with similar usage conditions are reassembled into batteries for echelon utilization or other small electric motorcycles; if the SOH value is lower than 50%, they need to be disassembled entirely [16]. Unfortunately, the SOH value of the battery on an electric vehicle cannot be measured directly. SOH estimation is closely related to the battery usage history. Standard prediction methods are based on capacity, resistance, voltage, current, and temperature [32]. The traditional measurement method removes the battery and measures the internal resistance. Although some non-destructive testing methods have been proposed, such as ultrasonic testing and X-ray scanning, the accuracy of these methods is generally low [33]. Since AI has advantages in data processing, it is

widely used in battery SOH estimation. In recent years, there have been some reviews on EVB SOH estimation, but they have focused more on the underlying estimation methods and algorithms. Yang et al. [34] discussed various SOH estimation methods based on capacity, impedance, and aging mechanism parameters, analyzed current SOH diagnosis and prediction challenges, and discussed future development trends. Manoharan et al. [35] reviewed the application of artificial neural networks, gradient boosting, and support vector machines for electric vehicle battery state of charge (SOC) and SOH estimation. Li et al. [36] reviewed the development of a remaining service life (RUL) prediction for lithium-ion batteries based on machine learning. They discussed the application of different algorithms, including recurrent neural networks (RNNs) and support vector machines (SVMs). This section compares characterization parameters and algorithms in SOH estimation, analyzing their characteristics, advantages, and disadvantages.

3.1. AI Methods in SOH Estimation

AI-based methods, such as transfer learning, reinforcement learning, neural networks, etc., are widely used in EVB SOH estimation. AI-based battery SOH estimation usually involves offline training and online estimation. During training, informative health indicators are extracted from aging data. The AI-based model then learns and updates its weights and biases to fit this training data. After the model is well trained, it is applied to the battery management system (BMS) to estimate the real SOH based on raw data [37]. Ruan et al. [38] proposed a lithium-ion battery health-state estimation method based on a convolutional neural network and transfer learning, using data from the constant-current and constantvoltage charging conversion stage to achieve high-precision and robust SOH prediction. Li et al. [39] proposed a health-conscious battery state estimation method based on deep transfer learning, which collects and processes battery operating data through a cloud platform, significantly improving the accuracy and stability of different electric vehicle battery models. Li et al. [40] proposed an end-to-end framework based on a hybrid neural network and Bayesian optimization for SOH estimation and the RUL prediction of lithium-ion batteries for electric vehicles, verifying its high accuracy and robustness. Couture et al. [32] proposed a transfer learning hybrid model that combines images and health indicators, experimentally verified its efficiency and accuracy in battery SOH estimation and RUL prediction, and demonstrated the potential of image data in regression tasks. AI-based methods used in the SOH process include random forest (RF), long short-term memory (LSTM), transfer learning (TL), support vector machine (SVM), and convolutional neural networks (CNNs). Table 3 describes the methods and their advantages and disadvantages.

AI Method	Description	Advantages	Disadvantages	Ref.
RF	A powerful and flexible ensemble learning method that improves model accuracy and robustness by combining multiple decision trees.	 High prediction accuracy and generalization ability. Reducing the risk of model overfitting. Can evaluate the importance of individual features. High robustness. 	 High computational complexity. Poor model interpretability. Not suitable for real-time predictions. 	[41–44]
LSTM	A recurrent neural network capable of learning and remembering long-term dependencies; often used to process sequence data such as time series analysis and natural language processing.	 Suitable for processing time series data. Solving the vanishing gradient problem 	Long training time.Requires many data.High complexity.	[45–49]

AI Method	Description	Advantages	Disadvantages	Ref.
TL	Using knowledge learned from one task (the source task) to aid the learning process in another related but different task (the target task).	Low data requirements.Speeding up the training process.	 Requires dependencies between source and target tasks. Hard to turn pre-trained models. 	[39,50–57]
SVM	A powerful and flexible supervised learning algorithm that handles linear and nonlinear classification tasks by maximizing inter-class margins and using kernel functions.	 Suitable for data with higher dimensions. Strong robustness. Able to handle non-linear classification problems 	 High computational complexity. Sensitive to parameter selection. Sensitive to missing data. 	[58–61]
CNN	A powerful deep learning model that automatically extracts features through convolutional and pooling layers and that is widely used in tasks such as image classification, target detection, and image segmentation.	 Can automatically extract features. Parameters can be shared. Preserving the spatial relationship of local features. 	 High computing resource requirements. Requiring many data. Sensitive to input size. 	[62–67]

Table 3. Cont.

3.2. Characterization Parameters in SOH Estimation

In EVB SOH estimation, commonly selected characteristic parameters include capacity, temperature, impedance, etc. Battery capacity is a crucial parameter describing the battery's SOH. In a laboratory, capacity loss can be measured directly by charging or discharging a battery at a nominal current until the cut-off voltage is reached. However, performing this process in practice is challenging due to device and battery operating range limitations. He et al. [68] proposed a revised Lorentzian function-based voltage-capacity (RL-VC) model. New features of interest (FOIs) are extracted from constant-current charging data by fitting the RL-VC model. They determined that the FOI highly correlated with battery capacity through correlation analysis and calibrated the linear model for SOH estimation. Galeotti et al. [69] discussed lithium polymer batteries' performance and SOH estimation through electrochemical impedance spectroscopy (EIS) techniques. Experimental results show that the ohmic resistance of the battery increases with aging, and SOH can be evaluated through the relationship between ohmic resistance and available capacity. When a lithium-ion battery operates, it generates or absorbs heat, leading to fluctuations in temperature. Chen et al. [70] proposed a SOH estimation method based on temperature prediction and a gated recurrent unit (GRU) neural network. They extracted multi-dimensional health features from differential temperature profiles to reflect multiple aspects of battery degradation. Table 4 shows the advantages and disadvantages of these characteristics. These AI-based estimation methods can achieve high accuracy. However, it should be noted that these prediction methods can sometimes only serve as auxiliary decision-making and cannot replace the actual detection process. When there are many battery cells, and the usage of different battery cells is different, heterogeneity is likely to occur. In this case, AI-based detection methods may not be effective.

Parameter	Description	Advantages	Disadvantages	Ref.
Capacity	 The amount of electricity a battery can store when fully charged. Usually expressed in ampere-hours (Ah). 	 Directly reflects battery SOH. The estimation is the most accurate. 	• Requires a complete charge-discharge cycle, time-consuming, not suitable for real-time monitoring	[68,71–79]
Impedance	• The battery's resistance to alternating current, including ohmic resistance, polarization resistance, etc.	 Highly sensitive to changes in the internal state of the battery. Can detect minor aging characteristics. 	 Complex measurement and data processing. Requires specialized equipment and extensive electrochemical knowledge. 	[69,80–87]
Temperature	• Temperature changes caused by thermal effects during battery charging and discharging	• Can be monitored in real-time.	Highly influenced by environmental factors.Indirectly reflect the SOH of the battery.	[70,88,89]

 Table 4. Summary of characteristic parameters for SOH estimation.

4. Disassembly Sequence Planning (DSP) for EV Batteries

Disassembly process modeling and AI algorithms are two primary considerations in AI-driven disassembly sequence planning (DSP). The traditional approach involves disassembly engineers manually formulating a disassembly sequence based on years of experience and technical manuals. This method relies on personal experience and is relatively flexible, but it is prone to subjective bias, and it is less efficient when faced with complex disassembly tasks.

4.1. Disassembly Process Modeling

Disassembly representation and modeling are critical for efficient disassembly sequences and operations. Several factors influence the feasibility of a disassembly sequence, including the relationship between components, the constraints involved in disassembling the components, the geometry of the product, hazardous characteristics of disassembly, the tools required for the operation, and the components needed to disassemble the target component. The feasibility of the disassembly sequence depends mainly on the product's structure. How the product structure is represented directly affects the efficiency of the disassembly sequence search [90]. The main modeling methods are AND/OR graph modeling, Petri Net (PNt) modeling, and matrix-based modeling [18].

4.2. AI Algorithms in DSP

AI methods are widely studied in DSP. Xiao et al. [91] proposed a human-machine collaborative disassembly optimization method based on multi-agent reinforcement learning, which improved disassembly efficiency and safety by optimizing the disassembly sequence and task allocation. Hartono et al. [92] proposed a method to optimize robot disassembly plans using a bee algorithm to maximize profits, save energy, reduce the environmental impact, and achieve the automation and efficiency of the disassembly sequence in the remanufacturing process. Gao et al. [93] proposed a multi-agent strategy optimization method based on partially observable deep reinforcement learning to improve the efficiency and safety of human-machine collaboration in order to dismantle scrapped electric vehicle batteries. Allagui et al. [94] proposed a reinforcement learning-based disassembly sequence planning optimization method to improve disassembly efficiency and reduce costs by reducing tool and direction changes. Chu et al. [95] proposed a human-machine collaborative disassembly optimization method based on hybrid particle swarm optimization and Q-learning algorithms to improve the efficiency and safety of EOL EVB disassembly. The authors also developed some algorithms for disassembly planning on other products. These algorithms can also be considered for battery disassembly. A summary of these methods is given in Table 5.

Table 5. Summary of EVB disassembly sequence planning models.

Methods		Description	Applications	Ref.
Machine learning	Dynamic Bayesian network	• Calculate and compare different numbers of observation sequences through reasoning and observation sequences to verify the possibility of finding the optimal disassembly sequence.	EV battery	Xiao et al. [96]
	Q-network	• Learn optimal decisions by interacting with the environment to maximize cumulative rewards.	No mention	Allagui et al. [94]
	Q-learning	 Learn optimal policies by updating state–action value functions. The reward function is based on the disassembly time and target component status. 	Smartphone	Chen et al. [97]
	Multi-agent reinforcement learning	 Improve disassembly efficiency and safety by dynamically adjusting disassembly task assignments and paths. Multiple agents can work together in a shared environment. 	EV battery	Xiao et al. [91]
Metaheuristics optimization algorithm	Bees algorithm	• A swarm intelligence optimization algorithm based on bees' foraging behavior.	Gear pump	Hartono et al. [92]
	Artificial Bee Colony Algorithm	• A mathematical model of the random disassembly line balancing problem based on expected returns is proposed.	Cell phone	Guo et al. [98]
	Genetic algorithm	• The optimal solution to the problem is found by simulating the genetic and selection mechanisms in biological evolution.	CRT TV	Wang et al. [99]
	Particle swarm optimization	• Combining the particle swarm optimization algorithm enhances global search and local search capabilities.	EV battery	Chu et al. [95]

Compared with traditional electric vehicle battery disassembly sequence planning, AI-based methods can learn and continuously optimize the strategy and improve the efficiency and accuracy of subsequent disassembly tasks. Traditional methods often rely on manual experience, while AI-based systems can automatically generate optimal disassembly strategies through large amounts of data and learning algorithms, reducing the dependence on expert knowledge. However, the performance of AI-based systems relies on large amounts of high-quality data. If the data are insufficient or noisy, the accuracy

of the model may be affected, leading to unsatisfactory disassembly results. In an actual production environment, obtaining enough relevant data can be challenging, especially when dealing with new or rapidly changing battery designs. Complex AI models, especially deep learning models, usually require high-performance computing resources to train and run, which may lead to high implementation costs and time overhead.

5. AI-Driven Disassembly Operation for EV Batteries

This section discusses some AI-driven disassembly operations for electric vehicle battery disassembly.

5.1. Object and Defect Identification

The disassembly process is more complex than the assembly process, and it requires the real-time judgment of the status of the disassembly process and the product being disassembled. Target recognition plays a vital role in the disassembly process. Zhang et al. [100] used the YOLOv4 algorithm to detect screws during the disassembly process. YOLOv4 is an efficient single-stage target detection algorithm that combines advanced network architecture and data enhancement technology to achieve high-precision target detection while maintaining real-time detection speed [101]. Foo et al. [102] proposed a practical learning framework and showed how the system can learn relevant disassembly information for LCDs. After training, the system's success rate in identifying LCD parts increased significantly from 11% to 87%. Zheng et al. [103] used PointNet DNN to identify 12 parts of car engine turbochargers. This method generates point cloud data from a CAD model and simulates sensor data with different accuracy levels through a depth camera simulator for training. Foo et al. [104] proposed a method that combines image preprocessing, deep learning models, and ontological reasoning to improve the accuracy and efficiency of screw detection in automated e-waste disassembly. Li et al. [105] proposed an accurate screw detection method based on Faster R-CNN and an innovative rotating edge similarity (RES) algorithm, aiming to automatically disassemble screws in electronic scrap, especially screws on mobile phone motherboards.

AI can also be used to identify defects in components. The traditional defect recognition method is for operators to identify product defects through visual inspection. This method relies on the experience and skills of personnel. Although it is highly flexible, it is easily affected by subjective factors, such as fatigue, resulting in false detection or missed detection. AI, especially deep learning models, can automatically extract complex features from large amounts of images or data and identify subtle and complex defects. Compared with traditional methods that rely on manually set rules and features, AI can capture the details of defects more accurately. Deep learning models outperform traditional computer vision algorithms in terms of accuracy and processing time, which has led to their widespread use in defect detection [106]. Tabernik et al. [107] proposed a two-stage deep learning architecture, including a segmentation network and a decision network. This design allows the model to be trained using fewer training samples, which is suitable for situations where defect samples are limited in practical applications. Medak et al. [108] solved the defect detection problem in ultrasound images by introducing the EfficientDet deep learning architecture, demonstrating its great potential and superiority in practical applications. Zhang et al. [109] proposed a semi-supervised learning method based on generative adversarial networks (GANs), which effectively improves the performance of automatically detecting and segmenting image surface defects while reducing the reliance on large amounts of annotated data.

However, AI models, especially deep learning models, require many annotated data for training. For defect recognition, this means collecting and annotating many images or data sets containing different types of defects. This data acquisition and annotation process is time-consuming and expensive, especially when defect samples are scarce or difficult to collect. At the same time, AI models are susceptible to data quality. If there is noise, mislabeling, or a data imbalance in the training data (too few samples of certain defect categories), it may affect the performance of the model and lead to inaccurate recognition results.

5.2. Intelligent Tool Selection and Disassembly Line Balancing

During the disassembly process, appropriate tools should be selected based on the type of connections. The connection methods of electric vehicle batteries include threaded connections, glue connections, welding connections, etc. Different connection methods correspond to different disassembly methods. AI can improve the efficiency of tool selection. Wang et al. [110] proposed a tool selection model based on a genetic algorithm (GA) to evaluate the suitability and matching value of disassembly tools in order to select the best disassembly tools. Liang et al. [111] constructed a mixed-integer, non-linear programming (MINLP) model of the multi-objective partial disassembly and line balancing problem (PDLBP) to achieve the minimization of the four optimization objectives of the number of workstations, workstation load, tool switching times, and energy consumption.

Disassembly line balancing (DLB) refers to the reasonable allocation and arrangement of disassembly tasks during the product disassembly process to improve disassembly efficiency, maximize economic benefits, reduce energy consumption, and balance the load of each station [112]. Traditional methods usually rely on engineers' experience and knowledge to manually configure and adjust the production line. Engineers assign tasks to different workstations based on the complexity of the task, process time, and resource availability. Engineers often use heuristic methods, such as the "longest processing time first" or the "shortest processing time first", to assign tasks through simplified rules in order to try to balance the working time of each workstation. AI can automatically analyze the tasks and resources of a production line, intelligently assign tasks to different workstations, and reduce the reliance on human intervention. This improves design efficiency, especially when faced with complex production lines. AI can quickly generate optimization solutions. At the same time, it can monitor the operating status of a production line in real time, dynamically adjust task allocation based on real-time data, and optimize the balance of a production line. This enables the production line to respond quickly to environmental changes, demand fluctuations, or failures. Ren et al. [113] proposed a mathematical model to solve the bi-objective disassembly line scheduling problem (Bi-DLSP) in order to optimize the total disassembly time and smoothing exponent. Wang et al. [114] applied the genetic simulated annealing algorithm to the disassembly line balancing problem. Yin et al. [115] used the Pareto-discrete hummingbird algorithm to address the disassembly line balancing problem. Experimental results showed that this method is more efficient in solving problems than the Discrete Artificial Bee Colony Algorithm (DABC), Hybrid Genetic Algorithm (HGA), Ant Colony Optimization Algorithm (ACO), and Hybrid Artificial Bee Colony Algorithm (HABC).

5.3. Intelligent Separation Optimization

During the battery disassembly process, the casing and module must be separated. Standard methods include mechanical cutting, laser cutting, hydraulic shearing, and manual disassembly. AI technology has great potential in modeling and optimizing laser beam processing quality characteristics, including geometric characteristics, metallurgical characteristics, surface quality, and the material removal rate [116]. Ding et al. [117] proposed an ensemble model based on a generalized regression neural network (GRNN) and Nondominated Sorting Genetic Algorithm II (NSGA-II), which could be used to predict and optimize the quality characteristics of the fiber laser cutting of stainless steel. Pimenov et al. [118] reviewed modern approaches to cutting tool condition monitoring, particularly the application of sensors and AI technologies, demonstrating the potential of these technologies to improve machining accuracy, productivity, and tool life. Serin et al. [119] proposed collecting vibration, acoustic emission, current, and cutting force data through sensors and using deep learning methods for predictive and preventive maintenance.

This section discusses some possible directions for EVB disassembly.

6.1. Electric Vehicle Battery Swapping

Electric vehicle battery swapping stations are a new trend in electric vehicle charging. As shown in Figure 4 [120], when an electric vehicle is close to out of charge, the driver drives the car to a battery swapping station and directly replaces it with a fully charged battery, which is charged at the battery swapping station [121]. One of the drawbacks of electric cars compared to fuel cars is their weak range and long charging times. Even with fast charging, it takes about 30 min to reach 80% of full charge [122]. In contrast, the battery-swapping process takes about 3 min to reach a 100% state of charge (SOC) [123]. In addition to reducing the waiting time for electric vehicle users to charge, battery swapping stations can also independently plan the charging time, for example, charging the batteries at low peaks of electricity consumption, which helps reduce the economic cost of charging and maintain the stability of the power grid. Established in 2014, Nio has become a global intelligent EV company, with at least 30 battery swapping stations (BSSs) in Europe and 2200 stations globally.



Figure 4. Process of battery swapping (adapted from [121]).

While battery swapping stations have certain advantages, there are still some current challenges faced:

- Only batteries of the same type and size can be replaced. However, the batteries of different car manufacturers worldwide, and even models of the same manufacturer, are not the same. This makes it impossible for battery swapping stations that serve cars from other manufacturers to be interoperable.
- Battery manufacturing costs account for a large part of the cost of electric vehicle production. Establishing a battery swapping station means manufacturing many additional batteries, which will incur relatively high expenses.
- To popularize battery-swapping, battery-swapping stations need to be widely established, which requires more investment than battery fast-charging stations. It also requires manufacturers to be able to integrate the supply chain. Due to great difficulties in this area, Tesla shifted its research and development direction from battery swapping to fast charging [124].
- The number of battery swaps also has peaks and troughs over time. For example, people generally drive during holidays, and the frequency of battery swaps for electric vehicles is currently higher. This causes problems with the layout of a battery swapping station and the reserve of replaceable batteries. The economic benefits will be low if

there are too many idle batteries. If there are too few replaceable batteries, the user must wait to charge their replacement battery.

 Consumers' psychological factors must also be considered. Battery replacement means that a car component is constantly being replaced, and some consumers may doubt this operating mode.

The rational layout and management of battery replacement stations are crucial to optimizing service coverage, and there is currently a series of research on the distribution optimization of battery replacement stations. AI-based methods can analyze real-time data, such as vehicle location, driving behavior, battery status, and battery swapping station capacity to optimize battery allocation. By predicting vehicle needs, AI-based methods can prepare a suitable inventory, reduce the waiting time, and improve battery swapping efficiency. AI-based methods can monitor the operation of all battery swapping stations in real time and dynamically adjust battery allocation and scheduling strategies. For example, when demand in an area surges, AI-based methods can automatically dispatch more batteries and resources to that area to cope with the peak demand. Yang et al. [125] proposed a data-driven BSS location optimization model using a one-month GPS trajectory dataset containing 514 EVs. Wang et al. [126] developed deep learning methods to predict EV battery swapping demand in order to optimize BSS arrangement. Yang et al. [127] developed an optimal battery allocation model for BSSs of EVs. This technology can be given priority in some large-scale vehicles, such as taxis. This technology will be more promising if various electric vehicle manufacturers can unify battery charging and discharging power and size for different models.

6.2. Intelligent Design for Disassembly (DFD)

Design for disassembly (DFD) simplifies the disassembly process [128]. DFD is a green manufacturing concept in which products are designed for ease of disassembly to recover valuable reusable materials and components and simplify maintenance through costeffective separation. Therefore, by allowing the reuse, remanufacturing, and recycling of products, waste is reduced at the product's EOL. Traditional DFD usually adopts modular design, using standardized components and interfaces, so that different modules can be quickly identified and separated when the product is disassembled. Such a design simplifies the disassembly process and reduces the risk of damage to components. Based on traditional DFD technology, AI-based methods can help automatically generate optimized disassembly design solutions. At the same time, AI-based methods can take into account multiple design goals, such as minimizing the disassembly time, maximizing the material recovery rate, minimizing the cost, and providing optimal design suggestions to help designers find the best balance under multiple constraints. AI-based methods can also simulate the disassembly process through virtual simulation, predict problems that may be encountered in actual disassembly, and make improvements in the design stage. This predictive ability helps reduce design defects and improve product disassembly.

As shown in Figure 5, multiple intelligent DFD methods could be adopted. RFID and QR codes could be adopted during the manufacturing process to record information about EVBs, while processors and sensors could be placed inside the EVBs to monitor SOH conditions and estimate their RUL.

DFD is applied to make the components in a battery, such as the positive electrode, negative electrode, and electrolyte, more accessible for separation and recycling. Currently, recycling efforts are focused on cathode materials because these are the most economically valuable among retired LIBs [129]. Due to the graphite anode, an additional separation step is required. Anode-free batteries are a new trend that simplifies the disassembly process. Meanwhile, the anode-free battery, which removes excess lithium and combines a fully lithiated cathode with an uncoated current collector, can achieve the highest possible energy density [130]. Furthermore, this configuration saves costs, energy, and technical requirements associated with anode production, including slurry preparation, coating, and drying processes in the drying chamber [131]. The connection between each part can

also be optimized to achieve non-destructive disassembly. Connected components and modules can be manufactured using shape-memory polymers, which offer the advantages of industrial feasibility, morphological diversity, and synthetic flexibility [132].



Figure 5. Multiple intelligent DFD methods.

6.3. Digital Twin and Human–Robot Collaboration Applications

The remote operation of the disassembly process can significantly improve its safety. Since toxic gases may be released during the battery disassembly process, improper operation may cause the risk of combustion and explosion. Digital twin (DT) and human–robot collaboration (HRC) are useful tools for realizing remote disassembly.

As shown in Figure 6, DT is achieved through a physical system with a virtual representation [133,134], and the digital model mirrors the physical system. Mirroring capabilities can be achieved through a data exchange, which requires installing sensors on the physical system to collect and transmit data over the network to the virtual model. At the same time, physical entities can also be operated using virtual models. Augmented reality (AR) is an excellent medium for achieving this function. AR is a technology that overlays computergenerated information or images onto the real world [135]. AR technology can combine virtual information with the natural environment using cameras, sensors, computers, and display devices. The use of DT and AR can realize remote control and reduce the difficulty of control, significantly reducing the safety risks of the working environment. It should be noted that the virtual model in the DT and AR needs to simulate the operating robot arm and the workpiece to be processed and the environment. Otherwise, the control that can be achieved on the virtual side may be restricted due to the operating environment on the physical side. In the past, the virtual end of digital twins generally used manual modeling; however, with the development of AI technology and the emergence of 3D point cloud technology, virtual models can be quickly modeled by scanning entities [136]. DT relies on a large number of sensors and data collection devices, which may involve the processing of sensitive information.

Due to the great difficulty of disassembling electric vehicle batteries and the small operating space in part of the disassembly process, which makes it difficult for the robotic arm to operate, it is difficult to automate the disassembly process [17] entirely. Human–robot collaboration (HRC) provides new ideas for disassembly by combining human intelligence and decision-making with the strength and precision of robots. For example, complex disassembly tasks are better suited to manual operations, while robots better handle hazardous and repetitive tasks. In traditional HRC, robots usually perform repetitive and precision-demanding tasks, while human workers are responsible for more complex tasks that require flexibility. The tasks in HRC disassembly are usually predefined, with little room for dynamic adjustment. AI-based methods can dynamically adjust task allocation based on real-time task complexity, worker status, robot capabilities, and other factors to ensure optimal collaboration between humans and robots. This dynamic allocation removes the limitations of fixed task allocation in traditional HRC. At the same time, AI-based methods give robots the ability to understand and respond to human natural language commands and gestures, making the interaction between humans and machines more intuitive and natural. Yuan et al. [137] proposed a new heuristic algorithm based on a multicriteria assessment of human-robot collaboration. The proposed disassembly elasticity assessment method uses the fuzzy Bayesian-ANP-extension cloud model to convert human judgment into numerical values in order to help managers make better decisions. Guo et al. [138] proposed HRC partial destructive disassembly sequence planning (DSP) driven by multiple failures. The disassembly sequence is optimized through an improved genetic algorithm to improve the efficiency and automation of end-of-life product disassembly. Gao et al. [93] proposed HRC disassembly strategy optimization based on deep reinforcement learning. This approach enables each agent to choose a strategy that maximizes the overall gain, ensuring that humans and robots adopt optimal disassembly strategies. The chal-

lenge is that human workers may lack trust in AI-driven robots, especially when the AI's decision-making process is not transparent. Establishing and maintaining trust between humans and machines is an important challenge that needs to be achieved through reliable behavior and transparent decision-making processes.



Figure 6. Digital twin implementation process (Adapted from [133]).

As shown in Figure 7, disassembly tasks' classification is performed to determine the disassembly sequence for batteries and the characteristics of parts in batteries. The second step is to allocate disassembly resources to determine the tasks allocation in HRC. After that, the entire process will be integrated into a solution, and the solution will be evaluated.



Figure 7. Flow chart of human-robot collaboration (adapted from [95]).

Since humans and robots will operate in the same environment, the robot may harm humans during movement, and safety factors must be considered during operation. Xu

et al. [139] proposed a strategic disassembly information model. This method used the improved discrete bee algorithm to solve the HRC disassembly line balancing problem. Liu et al. [140] proposed a collision-free HRC system based on context awareness. The system can plan the robot's path to avoid collisions with human operators while reaching its target location promptly and recognize human operator gestures with low computational overhead, further improving assembly efficiency. Wang et al. [141] proposed a deep learning-enhanced DT framework in HRC manufacturing. The proposed framework significantly improves the security and reliability of the HRC system through deep learning technology.

6.4. Charging Optimization

Changes in the voltage, current, temperature, etc. during charging will also affect battery life. Improving charging parameters to extend battery life saves energy and protects the environment. At the same time, compared with fuel vehicles, electric vehicles have significant disadvantages, namely a slow charging time and low endurance. Electric cars will be further promoted if the charging rate can be improved.

Battery aging can be divided into cyclic aging and calendar aging. Cyclic aging occurs during charging and discharging, while calendar aging occurs when a battery rests [142]. The temperature during charging, including the ambient temperature and the heat generated during battery charging, impacts the battery life. Park et al. [142] proposed a model and showed that indoor charging stations can reduce battery aging management costs by up to 13.31% compared with outdoor charging stations. Chung et al. [143] investigated battery maintenance during extended periods of inactivity. They designed an optimal charge profile that maintains battery status under ideal conditions to minimize degradation during idle periods while still meeting charging energy requirements.

There is currently much research on fast charging. Wang et al. [144] proposed a multi-stage charging strategy based on a fractional-order model using the Moth-Flame Optimization (MFO) algorithm. The test showed that each fitness function part's current stage number, cut-off voltage, and weight significantly affect charging performance. The fitness function should be weighted differently based on specific requirements. Jiang et al. [145] proposed a fast-charging design based on Bayesian optimization. They explored three acquisition functions (i.e., expected improvement, improvement probability, and lower confidence bound) to minimize the charging time for single-step and multi-step constant-current charging profiles. However, fast charging often accelerates battery aging. AI-based methods can provide an innovative approach to optimizing the balance between fast charging and reducing battery aging. The voltage and current of the traditional charging method are constant. An AI-based system can monitor the battery's charging status (such as voltage, temperature, charging current, etc.) in real time and dynamically adjust the charging rate based on these data to optimize charging efficiency and prevent the battery from overheating. At the same time, charging data can also be used for battery state-of-health estimation. However, the challenge is that existing charging hardware and infrastructure may not be fully compatible with the needs of AI-based systems. In order to achieve AI-driven fast charging, existing hardware may need to be upgraded or modified, which may involve high cost.

7. Conclusions

With the popularity of electric vehicles, disposal of retired electric vehicles is being considered. The traditional metal extraction method of crushing retired electric vehicle batteries destroys their structure and can only facilitate recycling of raw materials, which is inefficient. One optimization method is to conduct SOH estimation on electric vehicle batteries. Batteries with SOH values lower than 80% but higher than 50% can be used for echelon utilization. They are systematically disassembled if the SOH value is lower than 50%. AI has excellent potential in EV battery disassembly. To evaluate AI applications in the EVB disassembly process, this survey has provided a more systematic summary

of AI applications in EV battery disassembly, including SOH estimation, disassembly sequence planning, and disassembly operations. The article has also discussed promising development directions in battery recycling, including battery swapping, intelligent design for disassembly, DT and HRC, and charging optimization. Further research in these areas is needed. Overall, this review has provided a systematic summary of AI in EVB disassembly and pointed out possible directions for future research.

Author Contributions: Conceptualization, S.K.O., A.Y.C.N. and Z.A.; methodology, Z.A.; formal analysis, Z.A.; investigation, Z.A.; resources, S.K.O. and A.Y.C.N.; data curation, Z.A.; writing—original draft preparation, Z.A.; writing—review and editing, S.K.O. and A.Y.C.N.; supervision, S.K.O. and A.Y.C.N.; project administration, S.K.O. and A.Y.C.N. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflicts of interest.

Declaration of Generative AI and AI-Assisted Technologies in the Writing Process: The authors declare that no generative AI or AI-assisted technologies were used in the writing process.

Abbreviations

The following abbreviations are used in this manuscript:

ACO	Ant Colony Optimization Algorithm
AI	Artificial intelligence
AR	Augmented reality
BMS	Battery management system
BSSs	Battery swapping stations
CNN	Convolutional neural network
DABC	Discrete Artificial Bee Colony Algorithm
DFD	Design for disassembly
DNN	Deep neural network
DSP	Disassembly sequence planning
DT	Digital twin
EIS	Electrochemical impedance spectroscopy
EOL	End of life
EVB	Electric vehicle battery
EVs	Electric vehicles
GA	Genetic algorithm
GRNN	Generalized regression neural network
HABC	Hybrid Artificial Bee Colony Algorithm
HGA	Hybrid genetic algorithm
LIBs	Lithium-ion batteries
LSTM	Long short-term memory
MFO	Moth–Flame Optimization
NSGA-II	Non-Dominated Sorting Genetic Algorithm II
PNt	Petri Net
RES	Rotating edge similarity algorithm
RF	Random forest
RUL	Remaining useful life
SOC	State of charge
SOH	State of health
SVM	Support vector machine
TL	Transfer learning

References

- Executive Summary—Global EV Outlook 2024—Analysis. Available online: https://www.iea.org/reports/global-ev-outlook-20 24/executive-summary (accessed on 15 May 2024).
- 2. Free Icons and Stickers. Available online: https://www.flaticon.com/ (accessed on 8 August 2024).
- 3. Vecteezy. Available online: https://www.vecteezy.com/free-vector (accessed on 8 August 2024).
- 4. Pixabay. Available online: https://pixabay.com/ (accessed on 8 August 2024).
- 5. Reinhart, L.; Vrucak, D.; Woeste, R.; Lucas, H.; Rombach, E.; Friedrich, B.; Letmathe, P. Pyrometallurgical Recycling of Different Lithium-Ion Battery Cell Systems: Economic and Technical Analysis. *J. Clean. Prod.* **2023**, *416*, 137834. [CrossRef]
- 6. Wu, J.; Xiao, L.; Shen, L.; Ran, J.-J.; Zhong, H.; Zhu, Y.-R.; Chen, H. Recent Advancements in Hydrometallurgical Recycling Technologies of Spent Lithium-Ion Battery Cathode Materials. *Rare Met.* **2024**, *43*, 879–899. [CrossRef]
- Lai, X.; Huang, Y.; Deng, C.; Gu, H.; Han, X.; Zheng, Y.; Ouyang, M. Sorting, Regrouping, and Echelon Utilization of the Large-Scale Retired Lithium Batteries: A Critical Review. *Renew. Sustain. Energy Rev.* 2021, 146, 111162. [CrossRef]
- 8. Wang, Y.; Hu, F.; Wang, Y.; Guo, J.; Yang, Z.; Jiang, F. Revolutionizing the Afterlife of EV Batteries: A Comprehensive Guide to Echelon Utilization Technologies. *ChemElectroChem* **2024**, *11*, e202300666. [CrossRef]
- Explosion Hits CATL Battery Precursors Plant in China—Latest Market News. Available online: https://www.argusmedia. com/en/news-and-insights/latest-market-news/2175179-explosion-hits-catl-battery-precursors-plant-in-china (accessed on 15 May 2024).
- 10. Zhang, C.; Lu, Y. Study on Artificial Intelligence: The State of the Art and Future Prospects. J. Ind. Inf. Integr. 2021, 23, 100224. [CrossRef]
- 11. Silver, D.; Schrittwieser, J.; Simonyan, K.; Antonoglou, I.; Huang, A.; Guez, A.; Hubert, T.; Baker, L.; Lai, M.; Bolton, A.; et al. Mastering the Game of Go without Human Knowledge. *Nature* **2017**, *550*, 354–359. [CrossRef]
- 12. Harper, G.; Sommerville, R.; Kendrick, E.; Driscoll, L.; Slater, P.; Stolkin, R.; Walton, A.; Christensen, P.; Heidrich, O.; Lambert, S.; et al. Recycling Lithium-Ion Batteries from Electric Vehicles. *Nature* **2019**, *575*, 75–86. [CrossRef]
- Neumann, J.; Petranikova, M.; Meeus, M.; Gamarra, J.D.; Younesi, R.; Winter, M.; Nowak, S. Recycling of Lithium-Ion Batteries— Current State of the Art, Circular Economy, and Next Generation Recycling. *Adv. Energy Mater.* 2022, 12, 2102917. [CrossRef]
- 14. Meng, K.; Xu, G.; Peng, X.; Youcef-Toumi, K.; Li, J. Intelligent Disassembly of Electric-Vehicle Batteries: A Forward-Looking Overview. *Resour. Conserv. Recycl.* 2022, 182, 106207. [CrossRef]
- 15. Ji, H.; Wang, J.; Ma, J.; Cheng, H.-M.; Zhou, G. Fundamentals, Status and Challenges of Direct Recycling Technologies for Lithium Ion Batteries. *Chem. Soc. Rev.* 2023, 52, 8194–8244. [CrossRef]
- 16. Yu, W.; Guo, Y.; Xu, S.; Yang, Y.; Zhao, Y.; Zhang, J. Comprehensive Recycling of Lithium-Ion Batteries: Fundamentals, Pretreatment, and Perspectives. *Energy Storage Mater.* **2023**, *54*, 172–220. [CrossRef]
- 17. Gu, X.; Bai, H.; Cui, X.; Zhu, J.; Zhuang, W.; Li, Z.; Hu, X.; Song, Z. Challenges and Opportunities for Second-Life Batteries: Key Technologies and Economy. *Renew. Sustain. Energy Rev.* **2024**, *192*, 114191. [CrossRef]
- Li, W.; Peng, Y.; Zhu, Y.; Pham, D.T.; Nee, A.Y.C.; Ong, S.K. End-of-Life Electric Vehicle Battery Disassembly Enabled by Intelligent and Human-Robot Collaboration Technologies: A Review. *Rob. Comput. Integr. Manuf.* 2024, 89, 102758. [CrossRef]
- 19. Hertel, D.; Bräunig, G.; Thürer, M. Towards a Green Electromobility Transition: A Systematic Review of the State of the Art on Electric Vehicle Battery Systems Disassembly. *J. Manuf. Syst.* **2024**, *74*, 387–396. [CrossRef]
- 20. Vangapally, N.; Penki, T.R.; Elias, Y.; Muduli, S.; Maddukuri, S.; Luski, S.; Aurbach, D.; Martha, S.K. Lead-Acid Batteries and Lead–Carbon Hybrid Systems: A Review. *J. Power Sources* **2023**, *579*, 233312. [CrossRef]
- Zhou, D.; Guo, X.; Zhang, Q.; Shi, Y.; Zhang, H.; Yu, C.; Pang, H. Nickel-Based Materials for Advanced Rechargeable Batteries. *Adv. Funct. Mater.* 2022, 32, 2107928. [CrossRef]
- 22. Usiskin, R.; Lu, Y.; Popovic, J.; Law, M.; Balaya, P.; Hu, Y.-S.; Maier, J. Fundamentals, Status and Promise of Sodium-Based Batteries. *Nat. Rev. Mater.* 2021, *6*, 1020–1035. [CrossRef]
- Kim, T.; Song, W.; Son, D.Y.; Ono, L.K.; Qi, Y. Lithium-Ion Batteries: Outlook on Present, Future, and Hybridized Technologies. J. Mater. Chem. A 2019, 7, 2942–2964. [CrossRef]
- 24. Alfaro-Algaba, M.; Ramirez, F.J. Techno-Economic and Environmental Disassembly Planning of Lithium-Ion Electric Vehicle Battery Packs for Remanufacturing. *Resour. Conserv. Recycl.* **2020**, *154*, 104461. [CrossRef]
- Heilweil, R. The Race for a Better Battery Starts Now. Available online: https://www.vox.com/recode/23027110/solid-statelithium-battery-tesla-gm-ford (accessed on 17 May 2024).
- 26. A Bit About Batteries | Tesla Portugal. Available online: https://www.tesla.com/pt_pt/blog/bit-about-batteries (accessed on 8 July 2024).
- TDI Engine—Volkswagen Australia. Available online: https://www.volkswagen.com.au/en/hybrid-and-electric/technology/ id-technology/battery-technology.html (accessed on 8 July 2024).
- 28. BYD Blade Battery: The Turning Point Innovation in Electric Vehicle Batteries. 2024. Available online: https://www. bydbdautogroup.com/en/byd-blade-battery-en/ (accessed on 13 July 2024).
- 29. BYD's Revolutionary Blade Battery: All You Need to Know. Available online: https://www.byd.com/eu/blog/BYDs-revolutionary-Blade-Battery-all-you-need-to-know (accessed on 15 May 2024).
- 30. PRISMA Statement. Available online: https://www.prisma-statement.org (accessed on 8 August 2024).

- 31. What Are SOC and SOH of a Battery, How to Measure Them? Available online: https://www.biologic.net/topics/battery-statesstate-of-charge-soc-state-of-health-soh/ (accessed on 17 May 2024).
- 32. Couture, J.; Lin, X. Image- and Health Indicator-Based Transfer Learning Hybridization for Battery RUL Prediction. *Eng. Appl. Artif. Intell.* **2022**, *114*, 105120. [CrossRef]
- Pastor-Fernández, C.; Yu, T.F.; Widanage, W.D.; Marco, J. Critical Review of Non-Invasive Diagnosis Techniques for Quantification of Degradation Modes in Lithium-Ion Batteries. *Renew. Sustain. Energy Rev.* 2019, 109, 138–159. [CrossRef]
- Yang, S.; Zhang, C.; Jiang, J.; Zhang, W.; Zhang, L.; Wang, Y. Review on State-of-Health of Lithium-Ion Batteries: Characterizations, Estimations and Applications. J. Clean. Prod. 2021, 314, 128015. [CrossRef]
- Manoharan, A.; Begam, K.M.; Aparow, V.R.; Sooriamoorthy, D. Artificial Neural Networks, Gradient Boosting and Support Vector Machines for Electric Vehicle Battery State Estimation: A Review. J. Energy Storage 2022, 55, 105384. [CrossRef]
- Li, X.; Yu, D.; Søren Byg, V.; Daniel Ioan, S. The Development of Machine Learning-Based Remaining Useful Life Prediction for Lithium-Ion Batteries. J. Energy Chem. 2023, 82, 103–121. [CrossRef]
- Liu, K.; Wei, Z.; Zhang, C.; Shang, Y.; Teodorescu, R.; Han, Q.-L. Towards Long Lifetime Battery: AI-Based Manufacturing and Management. *IEEE/CAA J. Autom. Sin.* 2022, 9, 1139–1165. [CrossRef]
- Ruan, H.; Wei, Z.; Shang, W.; Wang, X.; He, H. Artificial Intelligence-Based Health Diagnostic of Lithium-Ion Battery Leveraging Transient Stage of Constant Current and Constant Voltage Charging. *Appl. Energy* 2023, 336, 120751. [CrossRef]
- Li, S.; He, H.; Zhao, P.; Cheng, S. Health-Conscious Vehicle Battery State Estimation Based on Deep Transfer Learning. *Appl. Energy* 2022, *316*, 119120. [CrossRef]
- 40. Li, P.; Zhang, Z.; Grosu, R.; Deng, Z.; Hou, J.; Rong, Y.; Wu, R. An End-to-End Neural Network Framework for State-of-Health Estimation and Remaining Useful Life Prediction of Electric Vehicle Lithium Batteries. *Renew. Sustain. Energy Rev.* 2022, 156, 111843. [CrossRef]
- Alfarizi, M.G.; Tajiani, B.; Vatn, J.; Yin, S. Optimized Random Forest Model for Remaining Useful Life Prediction of Experimental Bearings. *IEEE Trans. Ind. Inform.* 2023, 19, 7771–7779. [CrossRef]
- Li, Y.; Zou, C.; Berecibar, M.; Nanini-Maury, E.; Chan, J.C.-W.; van den Bossche, P.; Van Mierlo, J.; Omar, N. Random Forest Regression for Online Capacity Estimation of Lithium-Ion Batteries. *Appl. Energy* 2018, 232, 197–210. [CrossRef]
- Yang, N.; Song, Z.; Hofmann, H.; Sun, J. Robust State of Health Estimation of Lithium-Ion Batteries Using Convolutional Neural Network and Random Forest. J. Energy Storage 2022, 48, 103857. [CrossRef]
- 44. Mawonou, K.S.R.; Eddahech, A.; Dumur, D.; Beauvois, D.; Godoy, E. State-of-Health Estimators Coupled to a Random Forest Approach for Lithium-Ion Battery Aging Factor Ranking. *J. Power Sources* **2021**, *484*, 229154. [CrossRef]
- 45. Shu, X.; Zhang, Y.; Chen, Z.; Liu, Y. A Flexible State-of-Health Prediction Scheme for Lithium-Ion Battery Packs With Long Short-Term Memory Network and Transfer Learning. *IEEE Trans. Transp. Electrif.* **2021**, *7*, 2238–2248. [CrossRef]
- Wang, F.-K.; Amogne, Z.E.; Chou, J.-H.; Tseng, C. Online Remaining Useful Life Prediction of Lithium-Ion Batteries Using Bidirectional Long Short-Term Memory with Attention Mechanism. *Energy* 2022, 254, 124344. [CrossRef]
- Cheng, G.; Wang, X.; He, Y. Remaining Useful Life and State of Health Prediction for Lithium Batteries Based on Empirical Mode Decomposition and a Long and Short Memory Neural Network. *Energy* 2021, 232, 121022. [CrossRef]
- Cho, S.; Han, D.; Kim, J.; Kim, J. State of Health Estimation Embedded with Hardware Accelerator Based on Long Short-Term Memory Combined with Bayesian Optimization Considering Extracted Health Indicator in Charging Conditions. *J. Energy Storage* 2024, 90, 111897. [CrossRef]
- Li, P.; Zhang, Z.; Xiong, Q.; Ding, B.; Hou, J.; Luo, D.; Rong, Y.; Li, S. State-of-Health Estimation and Remaining Useful Life Prediction for the Lithium-Ion Battery Based on a Variant Long Short Term Memory Neural Network. *J. Power Sources* 2020, 459, 228069. [CrossRef]
- 50. Ma, G.; Xu, S.; Yang, T.; Du, Z.; Zhu, L.; Ding, H.; Yuan, Y. A Transfer Learning-Based Method for Personalized State of Health Estimation of Lithium-Ion Batteries. *IEEE Trans. Neural Netw. Learn. Syst.* **2024**, *35*, 759–769. [CrossRef]
- 51. Deng, Z.; Lin, X.; Cai, J.; Hu, X. Battery Health Estimation with Degradation Pattern Recognition and Transfer Learning. J. Power Sources 2022, 525, 231027. [CrossRef]
- 52. Ma, Y.; Shan, C.; Gao, J.; Chen, H. Multiple Health Indicators Fusion-Based Health Prognostic for Lithium-Ion Battery Using Transfer Learning and Hybrid Deep Learning Method. *Reliab. Eng. Syst. Saf.* **2023**, *229*, 108818. [CrossRef]
- 53. Che, Y.; Deng, Z.; Lin, X.; Hu, L.; Hu, X. Predictive Battery Health Management With Transfer Learning and Online Model Correction. *IEEE Trans. Veh. Technol.* **2021**, *70*, 1269–1277. [CrossRef]
- 54. Ma, G.; Xu, S.; Jiang, B.; Cheng, C.; Yang, X.; Shen, Y.; Yang, T.; Huang, Y.; Ding, H.; Yuan, Y. Real-Time Personalized Health Status Prediction of Lithium-Ion Batteries Using Deep Transfer Learning. *Energy Environ. Sci.* **2022**, *15*, 4083–4094. [CrossRef]
- 55. Ye, Z.; Yu, J. State-of-Health Estimation for Lithium-Ion Batteries Using Domain Adversarial Transfer Learning. *IEEE Trans. Power Electron.* 2022, 37, 3528–3543. [CrossRef]
- 56. Sahoo, S.; Hariharan, K.S.; Agarwal, S.; Swernath, S.B.; Bharti, R.; Han, S.; Lee, S. Transfer Learning Based Generalized Framework for State of Health Estimation of Li-Ion Cells. *Sci. Rep.* **2022**, *12*, 13173. [CrossRef] [PubMed]
- 57. Zhou, K.Q.; Qin, Y.; Yuen, C. Transfer-Learning-Based State-of-Health Estimation for Lithium-Ion Battery with Cycle Synchronization. *IEEE/ASME Trans. Mechatron.* 2023, 28, 692–702. [CrossRef]
- 58. Li, Q.; Li, D.; Zhao, K.; Wang, L.; Wang, K. State of Health Estimation of Lithium-Ion Battery Based on Improved Ant Lion Optimization and Support Vector Regression. *J. Energy Storage* 2022, *50*, 104215. [CrossRef]

- Yang, D.; Wang, Y.; Pan, R.; Chen, R.; Chen, Z. State-of-Health Estimation for the Lithium-Ion Battery Based on Support Vector Regression. *Appl. Energy* 2018, 227, 273–283. [CrossRef]
- Klass, V.; Behm, M.; Lindbergh, G. A Support Vector Machine-Based State-of-Health Estimation Method for Lithium-Ion Batteries under Electric Vehicle Operation. J. Power Sources 2014, 270, 262–272. [CrossRef]
- 61. Feng, X.; Weng, C.; He, X.; Han, X.; Lu, L.; Ren, D.; Ouyang, M. Online State-of-Health Estimation for Li-Ion Battery Using Partial Charging Segment Based on Support Vector Machine. *IEEE Trans. Veh. Technol.* **2019**, *68*, 8583–8592. [CrossRef]
- Gu, X.; See, K.W.; Li, P.; Shan, K.; Wang, Y.; Zhao, L.; Lim, K.C.; Zhang, N. A Novel State-of-Health Estimation for the Lithium-Ion Battery Using a Convolutional Neural Network and Transformer Model. *Energy* 2023, 262, 125501. [CrossRef]
- Xu, H.; Wu, L.; Xiong, S.; Li, W.; Garg, A.; Gao, L. An Improved CNN-LSTM Model-Based State-of-Health Estimation Approach for Lithium-Ion Batteries. *Energy* 2023, 276, 127585. [CrossRef]
- 64. Ouyang, T.; Su, Y.; Wang, C.; Jin, S. Combined Meta-Learning with CNN-LSTM Algorithms for State-of-Health Estimation of Lithium-Ion Battery. *IEEE Trans. Power Electron.* 2024, *39*, 10106–10117. [CrossRef]
- 65. Buchanan, S.; Crawford, C. Probabilistic Lithium-Ion Battery State-of-Health Prediction Using Convolutional Neural Networks and Gaussian Process Regression. *J. Energy Storage* 2024, *76*, 109799. [CrossRef]
- 66. Zhang, S.; Zhu, H.; Wu, J.; Chen, Z. Voltage Relaxation-Based State-of-Health Estimation of Lithium-Ion Batteries Using Convolutional Neural Networks and Transfer Learning. *J. Energy Storage* **2023**, *73*, 108579. [CrossRef]
- 67. Liu, B.; Xu, J.; Xia, W. State-of-Health Estimation for Lithium-Ion Battery Based on an Attention-Based CNN-GRU Model with Reconstructed Feature Series. *Int. J. Energy Res.* 2023, 2023, 8569161. [CrossRef]
- 68. He, J.; Wei, Z.; Bian, X.; Yan, F. State-of-Health Estimation of Lithium-Ion Batteries Using Incremental Capacity Analysis Based on Voltage–Capacity Model. *IEEE Trans. Transp. Electrific.* **2020**, *6*, 417–426. [CrossRef]
- Galeotti, M.; Cinà, L.; Giammanco, C.; Cordiner, S.; Di Carlo, A. Performance Analysis and SOH (State of Health) Evaluation of Lithium Polymer Batteries through Electrochemical Impedance Spectroscopy. *Energy* 2015, 89, 678–686. [CrossRef]
- 70. Chen, Z.; Zhao, H.; Zhang, Y.; Shen, S.; Shen, J.; Liu, Y. State of Health Estimation for Lithium-Ion Batteries Based on Temperature Prediction and Gated Recurrent Unit Neural Network. *J. Power Sources* **2022**, *521*, 230892. [CrossRef]
- She, C.; Zhang, L.; Wang, Z.; Sun, F.; Liu, P.; Song, C. Battery State-of-Health Estimation Based on Incremental Capacity Analysis Method: Synthesizing From Cell-Level Test to Real-World Application. *IEEE J. Emerg. Sel. Top. Power Electron.* 2023, 11, 214–223. [CrossRef]
- 72. Xu, Z.; Wang, J.; Lund, P.D.; Zhang, Y. Estimation and Prediction of State of Health of Electric Vehicle Batteries Using Discrete Incremental Capacity Analysis Based on Real Driving Data. *Energy* **2021**, *225*, 120160. [CrossRef]
- Chang, C.; Wang, Q.; Jiang, J.; Wu, T. Lithium-Ion Battery State of Health Estimation Using the Incremental Capacity and Wavelet Neural Networks with Genetic Algorithm. *J. Energy Storage* 2021, *38*, 102570. [CrossRef]
- Li, X.; Yuan, C.; Li, X.; Wang, Z. State of Health Estimation for Li-Ion Battery Using Incremental Capacity Analysis and Gaussian Process Regression. *Energy* 2020, 190, 116467. [CrossRef]
- Li, X.; Yuan, C.; Wang, Z. State of Health Estimation for Li-Ion Battery via Partial Incremental Capacity Analysis Based on Support Vector Regression. *Energy* 2020, 203, 117852. [CrossRef]
- 76. Zhang, Y.; Liu, Y.; Wang, J.; Zhang, T. State-of-Health Estimation for Lithium-Ion Batteries by Combining Model-Based Incremental Capacity Analysis with Support Vector Regression. *Energy* **2022**, *239*, 121986. [CrossRef]
- Zhang, S.; Zhai, B.; Guo, X.; Wang, K.; Peng, N.; Zhang, X. Synchronous Estimation of State of Health and Remaining Useful Lifetime for Lithium-Ion Battery Using the Incremental Capacity and Artificial Neural Networks. *J. Energy Storage* 2019, 26, 100951. [CrossRef]
- Guo, Y.; Yu, P.; Zhu, C.; Zhao, K.; Wang, L.; Wang, K. A State-of-health Estimation Method Considering Capacity Recovery of Lithium Batteries. Int. J. Energy Res. 2022, 46, 23730–23745. [CrossRef]
- 79. Bian, X.; Wei, Z.G.; Li, W.; Pou, J.; Sauer, D.U.; Liu, L. State-of-Health Estimation of Lithium-Ion Batteries by Fusing an Open-Circuit-Voltage Model and Incremental Capacity Analysis. *IEEE Trans. Power Electron.* **2022**, *37*, 2226–2236. [CrossRef]
- Jiang, B.; Zhu, J.; Wang, X.; Wei, X.; Shang, W.; Dai, H. A Comparative Study of Different Features Extracted from Electrochemical Impedance Spectroscopy in State of Health Estimation for Lithium-Ion Batteries. *Appl. Energy* 2022, 322, 119502. [CrossRef]
- Obregon, J.; Han, Y.-R.; Ho, C.W.; Mouraliraman, D.; Lee, C.W.; Jung, J.-Y. Convolutional Autoencoder-Based SOH Estimation of Lithium-Ion Batteries Using Electrochemical Impedance Spectroscopy. J. Energy Storage 2023, 60, 106680. [CrossRef]
- 82. Locorotondo, E.; Cultrera, V.; Pugi, L.; Berzi, L.; Pierini, M.; Lutzemberger, G. Development of a Battery Real-Time State of Health Diagnosis Based on Fast Impedance Measurements. *J. Energy Storage* **2021**, *38*, 102566. [CrossRef]
- 83. Messing, M.; Shoa, T.; Habibi, S. Estimating Battery State of Health Using Electrochemical Impedance Spectroscopy and the Relaxation Effect. *J. Energy Storage* **2021**, *43*, 103210. [CrossRef]
- 84. Yang, Q.; Xu, J.; Li, X.; Xu, D.; Cao, B. State-of-Health Estimation of Lithium-Ion Battery Based on Fractional Impedance Model and Interval Capacity. *Int. J. Electr. Power Energy Syst.* 2020, 119, 105883. [CrossRef]
- 85. Sun, X.; Zhang, Y.; Zhang, Y.; Wang, L.; Wang, K. Summary of Health-State Estimation of Lithium-Ion Batteries Based on Electrochemical Impedance Spectroscopy. *Energies* 2023, *16*, 5682. [CrossRef]
- Zhang, Q.; Huang, C.-G.; Li, H.; Feng, G.; Peng, W. Electrochemical Impedance Spectroscopy Based State-of-Health Estimation for Lithium-Ion Battery Considering Temperature and State-of-Charge Effect. *IEEE Trans. Transp. Electrific.* 2022, *8*, 4633–4645. [CrossRef]

- 87. Liu, Y.; Wang, L.; Li, D.; Wang, K. State-of-Health Estimation of Lithium-Ion Batteries Based on Electrochemical Impedance Spectroscopy: A Review. *Prot. Control. Mod. Power Syst.* 2023, *8*, 41. [CrossRef]
- Zhang, W.; He, H.; Li, T.; Yuan, J.; Xie, Y.; Long, Z. Lithium-Ion Battery State of Health Prognostication Employing Multi-Model Fusion Approach Based on Image Coding of Charging Voltage and Temperature Data. *Energy* 2024, 296, 131095. [CrossRef]
- Tian, J.; Xiong, R.; Shen, W. State-of-Health Estimation Based on Differential Temperature for Lithium Ion Batteries. *IEEE Trans.* Power Electron. 2020, 35, 10363–10373. [CrossRef]
- Ong, S.K.; Chang, M.M.L.; Nee, A.Y.C. Product Disassembly Sequence Planning: State-of-the-Art, Challenges, Opportunities and Future Directions. Int. J. Prod. Res. 2021, 59, 3493–3508. [CrossRef]
- Xiao, J.; Gao, J.; Anwer, N.; Eynard, B. Multi-Agent Reinforcement Learning Method for Disassembly Sequential Task Optimization Based on Human–Robot Collaborative Disassembly in Electric Vehicle Battery Recycling. J. Manuf. Sci. Eng. 2023, 145, 121001. [CrossRef]
- 92. Hartono, N.; Ramírez, F.J.; Pham, D.T. Optimisation of Robotic Disassembly Plans Using the Bees Algorithm. *Robot. Comput. Integr. Manuf.* 2022, *78*, 102411. [CrossRef]
- Gao, J.; Wang, G.; Xiao, J.; Zheng, P.; Pei, E. Partially Observable Deep Reinforcement Learning for Multi-Agent Strategy Optimization of Human-Robot Collaborative Disassembly: A Case of Retired Electric Vehicle Battery. *Robot. Comput. Integr. Manuf.* 2024, 89, 102775. [CrossRef]
- 94. Allagui, A.; Belhadj, I.; Plateaux, R.; Hammadi, M.; Penas, O.; Aifaoui, N. Reinforcement Learning for Disassembly Sequence Planning Optimization. *Comput. Ind.* 2023, 151, 103992. [CrossRef]
- 95. Chu, M.; Chen, W. Human-Robot Collaboration Disassembly Planning for End-of-Life Power Batteries. J. Manuf. Syst. 2023, 69, 271–291. [CrossRef]
- Xiao, J.; Anwer, N.; Li, W.; Eynard, B.; Zheng, C. Dynamic Bayesian Network-Based Disassembly Sequencing Optimization for Electric Vehicle Battery. CIRP J. Manuf. Sci. Technol. 2022, 38, 824–835. [CrossRef]
- 97. Chen, Z.; Li, L.; Zhao, F.; Sutherland, J.W.; Yin, F. Disassembly Sequence Planning for Target Parts of End-of-Life Smartphones Using Q-Learning Algorithm. *Procedia CIRP* 2023, *116*, 684–689. [CrossRef]
- Guo, H.; Zhang, L.; Ren, Y.; Li, Y.; Zhou, Z.; Wu, J. Optimizing a Stochastic Disassembly Line Balancing Problem with Task Failure via a Hybrid Variable Neighborhood Descent-Artificial Bee Colony Algorithm. *Int. J. Prod. Res.* 2023, *61*, 2307–2321. [CrossRef]
- Wang, K.; Guo, J.; Du, B.; Li, Y.; Tang, H.; Li, X.; Gao, L. A Novel MILP Model and an Improved Genetic Algorithm for Disassembly Line Balancing and Sequence Planning with Partial Destructive Mode. *Comput. Ind. Eng.* 2023, 186, 109704. [CrossRef]
- 100. Zhang, X.; Eltouny, K.; Liang, X.; Behdad, S. Automatic Screw Detection and Tool Recommendation System for Robotic Disassembly. J. Manuf. Sci. Eng. 2023, 145, 031008. [CrossRef]
- Terven, J.; Córdova-Esparza, D.-M.; Romero-González, J.-A. A Comprehensive Review of YOLO Architectures in Computer Vision: From YOLOv1 to YOLOv8 and YOLO-NAS. *Mach. Learn. Knowl. Extr.* 2023, 5, 1680–1716. [CrossRef]
- 102. Foo, G.; Kara, S.; Pagnucco, M. Artificial Learning for Part Identification in Robotic Disassembly Through Automatic Rule Generation in an Ontology. *IEEE Trans. Automat. Sci. Eng.* **2023**, *20*, 296–309. [CrossRef]
- 103. Zheng, S.; Lan, F.; Baronti, L.; Pham, D.T.; Castellani, M. Automatic Identification of Mechanical Parts for Robotic Disassembly Using the PointNet Deep Neural Network. *Int. J. Manuf. Res.* **2022**, *17*, 1–21. [CrossRef]
- Foo, G.; Kara, S.; Pagnucco, M. Screw Detection for Disassembly of Electronic Waste Using Reasoning and Re-Training of a Deep Learning Model. *Procedia CIRP* 2021, 98, 666–671. [CrossRef]
- 105. Li, X.; Li, M.; Wu, Y.; Zhou, D.; Liu, T.; Hao, F.; Yue, J.; Ma, Q. Accurate Screw Detection Method Based on Faster R-CNN and Rotation Edge Similarity for Automatic Screw Disassembly. *Int. J. Comput. Integr. Manuf.* 2021, 34, 1177–1195. [CrossRef]
- Tulbure, A.-A.; Tulbure, A.-A.; Dulf, E.-H. A Review on Modern Defect Detection Models Using DCNNs—Deep Convolutional Neural Networks. J. Adv. Res. 2022, 35, 33–48. [CrossRef]
- Tabernik, D.; Šela, S.; Skvarč, J.; Skočaj, D. Segmentation-Based Deep-Learning Approach for Surface-Defect Detection. J. Intell. Manuf. 2020, 31, 759–776. [CrossRef]
- Medak, D.; Posilovic, L.; Subasic, M.; Budimir, M.; Loncaric, S. Automated Defect Detection From Ultrasonic Images Using Deep Learning. *IEEE Trans. Ultrason. Ferroelectr. Freq. Control.* 2021, 68, 3126–3134. [CrossRef]
- Zhang, G.; Pan, Y.; Zhang, L. Semi-Supervised Learning with GAN for Automatic Defect Detection from Images. *Autom. Constr.* 2021, 128, 103764. [CrossRef]
- Wang, G.; Wu, H.; Xiao, J. A Genetic Algorithm-Based Optimization Approach for Disassembly Tool Selections toward Electric Vehicle Battery Disassembly. In Proceedings of the 2023 9th International Conference on Mechatronics and Robotics Engineering (ICMRE), Shenzhen, China, 10 February 2023; pp. 192–196.
- 111. Liang, W.; Zhang, Z.; Zhang, Y.; Xu, P.; Yin, T. Improved Social Spider Algorithm for Partial Disassembly Line Balancing Problem Considering the Energy Consumption Involved in Tool Switching. *Int. J. Prod. Res.* **2023**, *61*, 2250–2266. [CrossRef]
- 112. Özceylan, E.; Kalayci, C.B.; Güngör, A.; Gupta, S.M. Disassembly Line Balancing Problem: A Review of the State of the Art and Future Directions. *Int. J. Prod. Res.* **2019**, *57*, 4805–4827. [CrossRef]
- 113. Ren, Y.; Gao, K.; Fu, Y.; Li, D.; Suganthan, P.N. Ensemble Artificial Bee Colony Algorithm with Q-Learning for Scheduling Bi-Objective Disassembly Line. *Appl. Soft Comput.* **2024**, 155, 111415. [CrossRef]
- Wang, K.; Li, X.; Gao, L.; Li, P.; Gupta, S.M. A Genetic Simulated Annealing Algorithm for Parallel Partial Disassembly Line Balancing Problem. *Appl. Soft Comput.* 2021, 107, 107404. [CrossRef]

- 115. Yin, T.; Zhang, Z.; Jiang, J. A Pareto-Discrete Hummingbird Algorithm for Partial Sequence-Dependent Disassembly Line Balancing Problem Considering Tool Requirements. *J. Manuf. Syst.* **2021**, *60*, 406–428. [CrossRef]
- Bakhtiyari, A.N.; Wang, Z.; Wang, L.; Zheng, H. A Review on Applications of Artificial Intelligence in Modeling and Optimization of Laser Beam Machining. Opt. Laser Technol. 2021, 135, 106721. [CrossRef]
- 117. Ding, H.; Wang, Z.; Guo, Y. Multi-Objective Optimization of Fiber Laser Cutting Based on Generalized Regression Neural Network and Non-Dominated Sorting Genetic Algorithm. *Infrared Phys. Technol.* **2020**, *108*, 103337. [CrossRef]
- Pimenov, D.Y.; Bustillo, A.; Wojciechowski, S.; Sharma, V.S.; Gupta, M.K.; Kuntoğlu, M. Artificial Intelligence Systems for Tool Condition Monitoring in Machining: Analysis and Critical Review. J. Intell. Manuf. 2023, 34, 2079–2121. [CrossRef]
- Serin, G.; Sener, B.; Ozbayoglu, A.M.; Unver, H.O. Review of Tool Condition Monitoring in Machining and Opportunities for Deep Learning. Int. J. Adv. Manuf. Technol. 2020, 109, 953–974. [CrossRef]
- 120. Freepik Icon by Kiranshastry. Available online: https://www.freepik.com (accessed on 8 August 2024).
- 121. Wu, H. A Survey of Battery Swapping Stations for Electric Vehicles: Operation Modes and Decision Scenarios. *IEEE Trans. Intell. Transport. Syst.* **2022**, 23, 10163–10185. [CrossRef]
- Cui, D.; Wang, Z.; Liu, P.; Wang, S.; Dorrell, D.G.; Li, X.; Zhan, W. Operation Optimization Approaches of Electric Vehicle Battery Swapping and Charging Station: A Literature Review. *Energy* 2023, 263, 126095. [CrossRef]
- 123. NIO Power—NIO. Available online: https://www.nio.com/nio-power?noredirect= (accessed on 16 May 2024).
- 124. Valdes-Dapena, P. Tesla Failed at Battery Swapping but Stellantis Says It May Have the Secret; CNN Business. Available online: https://www.cnn.com/2023/12/08/business/tesla-battery-swapping-stellantis/index.html (accessed on 16 May 2024).
- 125. Yang, X.; Shao, C.; Zhuge, C.; Sun, M.; Wang, P.; Wang, S. Deploying Battery Swap Stations for Shared Electric Vehicles Using Trajectory Data. *Transp. Res. Part D Transp. Environ.* **2021**, *97*, 102943. [CrossRef]
- 126. Wang, S.; Chen, A.; Wang, P.; Zhuge, C. Short-Term Electric Vehicle Battery Swapping Demand Prediction: Deep Learning Methods. *Transp. Res. Part D Transp. Environ.* **2023**, *119*, 103746. [CrossRef]
- Yang, J.; Liu, W.; Ma, K.; Yue, Z.; Zhu, A.; Guo, S. An Optimal Battery Allocation Model for Battery Swapping Station of Electric Vehicles. *Energy* 2023, 272, 127109. [CrossRef]
- 128. Abuzied, H.; Senbel, H.; Awad, M.; Abbas, A. A Review of Advances in Design for Disassembly with Active Disassembly Applications. *Eng. Sci. Technol. Int. J.* **2020**, *23*, 618–624. [CrossRef]
- 129. Mao, J.; Ye, C.; Zhang, S.; Xie, F.; Zeng, R.; Davey, K.; Guo, Z.; Qiao, S. Toward Practical Lithium-Ion Battery Recycling: Adding Value, Tackling Circularity and Recycling-Oriented Design. *Energy Environ. Sci.* 2022, *15*, 2732–2752. [CrossRef]
- Nanda, S.; Gupta, A.; Manthiram, A. Anode-Free Full Cells: A Pathway to High-Energy Density Lithium-Metal Batteries. Adv. Energy Mater. 2021, 11, 2000804. [CrossRef]
- Heubner, C.; Maletti, S.; Auer, H.; Hüttl, J.; Voigt, K.; Lohrberg, O.; Nikolowski, K.; Partsch, M.; Michaelis, A. From Lithium-Metal toward Anode-Free Solid-State Batteries: Current Developments, Issues, and Challenges. *Adv. Funct. Mater.* 2021, 31, 2106608. [CrossRef]
- Lendlein, A.; Gould, O.E.C. Reprogrammable Recovery and Actuation Behaviour of Shape-Memory Polymers. *Nat. Rev. Mater.* 2019, 4, 116–133. [CrossRef]
- VanDerHorn, E.; Mahadevan, S. Digital Twin: Generalization, Characterization and Implementation. *Decis. Support Syst.* 2021, 145, 113524. [CrossRef]
- 134. Liu, Y.K.; Ong, S.K.; Nee, A.Y.C. State-of-the-Art Survey on Digital Twin Implementations. Adv. Manuf. 2022, 10, 1–23. [CrossRef]
- 135. Nee, A.Y.C.; Ong, S.K.; Chryssolouris, G.; Mourtzis, D. Augmented Reality Applications in Design and Manufacturing. *CIRP Ann.* **2012**, *61*, 657–679. [CrossRef]
- 136. Fei, B.; Yang, W.; Chen, W.; Li, Z.; Li, Y.; Ma, T.; Hu, X.; Ma, L. Comprehensive Review of Deep Learning-Based 3D Point Cloud Completion Processing and Analysis. *IEEE Trans. Intell. Transport. Syst.* **2022**, *23*, 22862–22883. [CrossRef]
- Yuan, G.; Liu, X.; Zhang, C.; Pham, D.T.; Li, Z. A New Heuristic Algorithm Based on Multi-Criteria Resilience Assessment of Human–Robot Collaboration Disassembly for Supporting Spent Lithium-Ion Battery Recycling. *Eng. Appl. Artif. Intell.* 2023, 126, 106878. [CrossRef]
- 138. Guo, L.; Zhang, Z.; Zhang, X. Human–Robot Collaborative Partial Destruction Disassembly Sequence Planning Method for End-of-Life Product Driven by Multi-Failures. *Adv. Eng. Inform.* **2023**, *55*, 101821. [CrossRef]
- Xu, W.; Cui, J.; Liu, B.; Liu, J.; Yao, B.; Zhou, Z. Human-Robot Collaborative Disassembly Line Balancing Considering the Safe Strategy in Remanufacturing. J. Clean. Prod. 2021, 324, 129158. [CrossRef]
- 140. Liu, H.; Wang, L. Collision-Free Human-Robot Collaboration Based on Context Awareness. *Robot. Comput. Integr. Manuf.* 2021, 67, 101997. [CrossRef]
- Wang, S.; Zhang, J.; Wang, P.; Law, J.; Calinescu, R.; Mihaylova, L. A Deep Learning-Enhanced Digital Twin Framework for Improving Safety and Reliability in Human–Robot Collaborative Manufacturing. *Robot. Comput. Integr. Manuf.* 2024, 85, 102608. [CrossRef]
- 142. Park, S.-W.; Son, S.-Y. Techno-Economic Analysis for the Electric Vehicle Battery Aging Management of Charge Point Operator. Energy 2023, 280, 128095. [CrossRef]
- 143. Chung, C.-H.; Jangra, S.; Lai, Q.; Lin, X. Optimization of Electric Vehicle Charging for Battery Maintenance and Degradation Management. *IEEE Trans. Transport. Electrific.* 2020, *6*, 958–969. [CrossRef]

- 144. Wang, Y.; Zhao, G.; Zhou, C.; Li, M.; Chen, Z. Lithium-Ion Battery Optimal Charging Using Moth-Flame Optimization Algorithm and Fractional-Order Model. *IEEE Trans. Transp. Electrific.* 2023, *9*, 4981–4989. [CrossRef]
- 145. Jiang, B.; Berliner, M.D.; Lai, K.; Asinger, P.A.; Zhao, H.; Herring, P.K.; Bazant, M.Z.; Braatz, R.D. Fast Charging Design for Lithium-Ion Batteries via Bayesian Optimization. *Appl. Energy* 2022, 307, 118244. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.